Author's response to reviews

Title: Psychosomatic syndromes and Anorexia Nervosa

Authors:

Giovanni Abbate-Daga (giovanni.abbatedaga@unito.it)
Nadia Delsedime (nadia.delsedime@unito.it)
Barbara Nicotra (barbara.nicotra@gmail.com)
Cristina Giovannone (cgiovannone@cottolengo.org)
Enrica Marzola (enrica.marzola@gmail.com)
Federico Amianto (federico.amianto@unito.it)
Secondo Fassino (secondo.fassino@unito.it)

Version: 2 Date: 25 May 2012

Author's response to reviews: see over
Dear Catherine Olino,

Please find attached our revised manuscript “Psychosomatic syndromes and Anorexia Nervosa” since we would like to have it evaluated for publication.

As requested, we emended the method section stating “All participants provided written informed consent according to the Ethical Committee of the Department of Neuroscience of the University of Turin”.

Thank you for your help to improve our manuscript.

Best regards,

Prof. Giovanni Abbate-Daga

Turin, 25th May 2012